Author's response to reviews

Title: Syphilis among people with HIV infection in southern Ethiopia: sero-prevalence and risk factors

Authors:

Techalew Shimelis (techalew03@yahoo.com)
Kinfe Lemma (wudiekinfe@yahoo.com)
Henock Ambachew (henock_ama@yahoo.com)
Endale Tadesse (endaletd@yahoo.com)

Version: 5 Date: 25 March 2015

Author's response to reviews:

We would like to thank the Editor for the valuable comments and we have accommodated the comments accordingly.

Comment 1: Please re-name Introduction section to Background section.
Response: We have revised it.

Comment 2: Please clarify if consent was written or verbal. Please state in the Methods section whether written informed consent for participation in the study was obtained from participants or, where participants are children, a parent or guardian.
Response: We have now clarified it.